Author's response to reviews

Title: Neutral sphingomyelinase 2 modulates cytotoxic effects of protopanaxadiol on different human cancer cells

Authors:

Bonggoo Park (parkbg35@gmail.com)
Yong-Moon Lee (ymleefn@chungbuk.ac.kr)
Jae-Sung Kim (jae.kim@surgery.ufl.edu)
Youl Her (marketing@btgin.com)
Ju Hee Kang (applekjh0503@hanmail.net)
Seung-Hyun Oh (eyeball@gachon.ac.kr)
Hwan-Mook Kim (hwanmook@gachon.ac.kr)

Version: 5 Date: 8 July 2013

Author's response to reviews:

Editorial comments: - Thank you for including further details on the contribution made by BTGin to your study in your Competing Interests section. However, we would still ask you to clearly declare in this section that the manuscript author Youl Her is affiliated with BTGin.

Response: As you have suggested in editorial comments, we have made changes in “Competing Interest” section by stating that one of the BTGin personnel, Youl Her, is included as an author in our manuscript.